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Began troubleshoot of 04 valve full open with cryo, valve checkd good. 

Heidi sugested check of LL meter, noted that the 30% offset was still there. Control point for the LL at cryo was 71%. with the offset all we could get to was 70.1%. 

Reset cryo set point to 68% and the valve closes, problem seems to be in the meter or the isolation module. 

During process the right dipole quenched, awaiting refill......
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